% begin preamble
\documentclass[11pt,titlepage]{article}
\usepackage[margin=1in,noheadfoot]{geometry}
\usepackage[parfill]{parskip}
\usepackage{graphicx}
\usepackage[colorlinks,urlcolor=blue]{hyperref}
\usepackage{amssymb}
\usepackage{amsbsy}
\usepackage{wrapfig}

\newcommand{\e}[1]{\times 10^{#1}}
\newcommand{\q}{\mathrm}
\newcommand{\where}[1][where]{\qquad\mbox{#1}\qquad}
\newcommand{\bs}[1]{\boldsymbol{#1}}
\newcommand{\bh}[1]{\hat{\boldsymbol{#1}}}
\newcommand{\bv}[1]{\vec{\boldsymbol{#1}}}
\newcommand{\bt}[1]{\tilde{\boldsymbol{#1}}}
\newcommand{\ad}{^{\dagger}}
\newcommand{\flow}[1]{\mathcal{E}_{#1}}
\newcommand{\flux}[1]{\Phi_{#1}}
\newcommand{\dslash}{d \hspace{-0.5ex}\rule[1.2ex]{0.6ex}{.1ex}}
\def\rcurs{{\mbox{$\resizebox{.16in}{.08in}{\includegraphics{ScriptR}}$}\!\!}}
\def\brcurs{{\mbox{$\resizebox{.16in}{.08in}{\includegraphics{BoldR}}$}\!\!}}
\def\hrcurs{{\mbox{$\hat \brcurs$}\!}}
%end preamble

\begin{document}
\begin{center}
  {\bf University of Kentucky, Physics 521 \\
    Homework \#14, Rev. A, due Monday, 2016-04-11 }
\end{center}

\vspace{.3 in}

{\bf 0.} Griffiths [2ed]
Ch.~7 \#1, %p298 gaussian trial functions
      \#4, %p298 delta in inf sq well
      \#5, %p298 perturbation overestimates
      \#6, %p303 ionization energy
%      \#8, %p308 direct,exchange integrals
      \#10, %p309 vibrations of H2 ion
      \#14, %p310 yukawa potential
      \#15, %p310 two-state system  LONG
      \#16, %p311 electron in magnetic field
      \#19. %p313 muon catalyzed fusion


\end{document}
